Title: Effectiveness of endometrial resection transcervical hysteroscopic based preventive treatment on the recurrence of endometrial polyps in post-menopausal women.

Authors:

Jesus S Jimenez-Lopez (jjimenez.hdoc@salud.madrid.org)
Ana Granada-Sanmigue (angranado@gmail.com)
Gregorio Lopez-Gonzalez (goyolopect2@hotmail.com)

Version: 2 Date: 16 September 2014

Author's response to reviews: see over
Dear Prof O’Donovani:

Through the electronic submission system http://mc.manuscriptcentral.com/jogr, we have uploaded the manuscript entitled “ENDOMETRIOSIS NODE IN GYNAECOLOGIC SCARS. A STUDY OF 17 PATIENTS AND THE DIAGNOSTIC CONSIDERATIONS IN CLINICAL EXPERIENCE IN A TERTIARY CARE CENTRE”, which is submitted to the consideration of the Editorial Board of the Journal for the section of Research articles

All authors have participated in the study to meet criteria of authorship, take responsibility for this study and have seen and approved the final draft. The manuscript has never published and is not being submitted simultaneously to another Journal.

We here report the results of a clinical series of 397 patients with endometrial polips treated and followed at a tertiary care hospital in Madrid, Spain. In the present study, we followed patients up to 60 months post-resection. In the initial procedure, the patients underwent hysteroscopic polypectomy, followed by endometrial resection. This study set out to determine the effectiveness of polypectomy hysteroscopic endometrial resection in preventing the recurrence of endometrial polyps in post-menopausal patients. We analyse the complications, additional surgery in patients and related patient satisfaction.

Thank you in advance for your attention to our contribution.

Yours sincerely,

Jesús S. Jiménez-López, MD, PhD

Service of Obstetrics and Gynecology
Hospital Universitario 12 de Octubre
Avenida de Córdoba s/n
E- 28041 Madrid, Spain
Tel. and fax: +34 91 3908376

E-mail: jjimenez.hdoc@salud.madrid.org and jjimenezme35426@gmail.com